Author's response to reviews

Title: The Impact of an Exercise Intervention on C - Reactive Protein During Pregnancy: a randomized controlled trial

Authors:

Marquis Hawkins (mshawkins@schoolph.umass.edu)
Barry Braun (Barry.Braun@colostate.edu)
Bess Marcus (bmarcus@ucsd.edu)
Edward Stanek III (stanek@schoolph.umass.edu)
Glenn Markenson (Glenn.Markenson@baystatehealth.org)
Lisa Chasan-Taber (lct@schoolph.umass.edu)

Version: 5
Date: 25 May 2015

Author's response to reviews:

May 25, 2015

Dr. Nawsheen Boodhun
Executive Editor
BMC Pregnancy and Childbirth

Dear Dr. Boodhun,

Thank you for your acceptance of our manuscript "The Impact of an Exercise Intervention on C - Reactive Protein During Pregnancy: a Randomized Controlled Trial," MS- 9050642771511736. The attached manuscript has been revised to include the editorial revisions.

Sincerely,

Marquis Hawkins
Assistant Professor of Epidemiology
Department of Biostatistics & Epidemiology
School of Public Health & Health Sciences
University of Massachusetts-Amherst